Week 28  

SM18 Operation SCHEDULE Meeting (Tuesdays)

 Bench status  and tests for this week : 


[image: image1.emf]BenchName Test type Temp --

TBA1 MBBR2048

Prep 5 Pumping + Cool 

Down Cooling

TBA2 MBBL1123 Thermal cycle Warming

TBB1 MBBL2077 PT 12 Warm Up Warming

TBB2 MBBR2065

Prep 5 Pumping + Cool 

Down Cooling

TBC1 MBAL1089 PT 12 Warm Up Warming

TBC2 SSS016

ICS 6 Disconnect Magnet 

WP07 300 K

TBD1 SSS020 PT 6 Training  1.9 K

TBD2 SSS017 Prep 2.2 IAP @ warm LF 300 K

TBE1 MBAL3182

PT 4 Slow power abort 

check 1.9 K

TBE2 MBBL1131

Prep 5 Pumping + Cool 

Down Cooling

TBF1 MBAL1069 EX1 Extra Test 1.9 K

TBF2 MBBL1115 ICS final connection



From Saturday 26/06-3/07 : 9 dipoles+ 1SSS has left the test bench.
· Magnets with Anticryostat: 2065 (special MM test), 2048 (repeated )
Test Coordinator this week:  G.De Rijk;  MTM  Tech. on call Richard:  (16-3602) . 

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

1. This week : more than 5 magnets in liquid phase- priority problems. 
2. Lost of cryo OK during MM (Saturday and Sunday)
· Magnets non conformties/ and or problems
1. 3143 (E2): YT122 circuit shorted to ground and dipole : detected at warm
2. 3132 (F1) : High peak voltage detected on the first quench : D2, D2_L, >250 V.HF protection circuits more efficient in AP2.
3. 2077 (B1): Step HV  Magnet/vs ground failed at 3.1 kV. IFS box problem.
4. 1069 (F1) : At the first training MIITS too high (39). High RRR?
5. SSS: Leads of EF circuit not isolated (SSS16),  wrong connection of octupole (SSS017) axis meas. for the sextupole is not working with SSSW2…

· AOB
1. New PTOFF version (AQA, hf off, lf off application) on sun MTA 9, 24,25,27,30,31,33,35,37,39 
2. Old remains active on sunias08, sun MTA 22,23,42,44.

	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_30_M
	TBC2
	A trigger was always present on lf acquisition.


	MTM
	lf
	Investigation performed by Maryline She found mistakes in the new config file of the recording of the QH discharge concerning the trigger definition. 

	Tuesday _30_N
	TBF1
	3132 magnet : Peak voltage exceeds the limit for D1, D2, D1_L, D_U, D2_U, D2_L (250 V)


	MTM
	AQA
	Training stopped. Investigation done by P.Pugnat. Origin was a bad coupling for one pair of the heaters.

OK to continue given by Pierre,

	Wednesday _30_M
	TBB1
	HV test at cold on 2077: failed first step at 3.1 kV


	ICS
	
	Special investigation launched. Dipole vs ground failed. IFS box dismounted and replaced. Then OK.

	Wednesday_31_N
	TBE2
	3143 magnet : IAP at warm shows error for LF discharge. HV test failed. YT122 shorted to the ground.


	
	
	Magnet non conform :disconnected.

	Wednesday_31_N
	TBA1
	2042 magnet : Alarm on AQA; Tau jump too high in ½ HF quench heater.
	MTM
	AQA
	Investigation by Pierre. Doubling of points and OK given by M.Buzio

	Thursday _01_M
	TBA1
	Diode test (PtE) repeated because connector “left” was put on “right” magnet. 
	MTM, 
	
	Test repeated without connector.

	Thursday _01_M
	TBC2
	SSS16: Problem with SSW measurement on the sextupole. The option “rotating wire “is not working with SSW Nº 2.

Not possible to power the octupole (circuit EF). 
	MTM
	
	Axis from sextupole not measured.

Waiting for an answer of J.Di Marco.

Investigation done by Salem at warm :inversion of the Vtaps cable+one lead of the EF circuit poorly isolated.

	Thursday _01_N

Sunday_04_N
	TBA2
	1023 magnet : Signal missing for Quench Antenna.

1023: Quench number 9: No quench location data
	MTM
	
	Guy informed. LQA setting has to be selected. 

ND: two settings are present in TEMA: QLA right and left depending on the polarity of the magnet.

	Saturday​_03​​​_A

Sunday_04_M
	TBB1


	LHC cycles on 2077: cryo OK lost 2 times.

Too many magnets between1.9 K and 4.2 K?

Priority problem?
	ACR/MTM
	
	NB: LHC cycle have to be repeated after a quench.

	Saturday​_03​​​_N
	TBD2


	IAP  at warm on SSS017: file stored in the folder of SSS037 because magnet name in TEMA was not changed.
	MTM
	
	

	Sunday​_0​​4_M
	TBA1


	Problem in signing the e-traveller for the 2048 magnet (repeat magnet) because the steps were signed for the first test.
	MTM
	TTS
	Traveller signed manually.

	Monday_05​​​_N
	TBD1


	SSS20. Not possible to power the EF corrector circuit.  
	MTM
	
	Inductance loop have to be changed from MQT to MO corrector.

	Monday_05​​​_N
	TBD2


	SSS17: Rvalues at warm not correct. 
	MTM
	
	Wrong connection: one  sextupole Vtaps not soldered., problem with one lead of EF circuit.

	Monday_05​​​_N
	TBF1


	1069 magnet : MIITS too high (39).  
	MTM
	
	Training stopped. P.Pugnat informed.

RRR measurements decided?
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		Bench		Name		Test type		Temp		-		-

		TBA1		MBBR2048		Prep 5 Pumping + Cool Down		Cooling

		TBA2		MBBL1123		Thermal cycle		Warming

		TBB1		MBBL2077		PT 12 Warm Up		Warming

		TBB2		MBBR2065		Prep 5 Pumping + Cool Down		Cooling

		TBC1		MBAL1089		PT 12 Warm Up		Warming

		TBC2		SSS016		ICS 6 Disconnect Magnet WP07		300 K

		TBD1		SSS020		PT 6 Training		1.9 K

		TBD2		SSS017		Prep 2.2 IAP @ warm LF		300 K

		TBE1		MBAL3182		PT 4 Slow power abort check		1.9 K

		TBE2		MBBL1131		Prep 5 Pumping + Cool Down		Cooling

		TBF1		MBAL1069		EX1 Extra Test		1.9 K

		TBF2		MBBL1115		ICS final connection






